Author's response to reviews

Title: Short term effects of a low-carbohydrate diet in overweight and obese subjects with low HDL-C levels

Authors:

Ahmet Selçuk Can (selcukcan@endokrinoloji.com)
Canan Uysal (cananuysal@superonline.com)
K. Erhan Palaoglu (kepalaoglu@amerikanhastanesi.org)

Version: 4 Date: 3 November 2010

Author's response to reviews:

Dear Editor:

Major Revisions:
The requested change was made. The sentence “All authors read and approved the final manuscript was added to Authors’ Contributions.

Minor Revisions:
Highlighting was removed.
Full citation details were provided for reference 6.
Full citation details were provided for reference 32.
Typographical errors were corrected.

Ahmet Selçuk Can, MD